Author’s response to reviews

Title: GALEGA OFFICINALIS EXTRACT REGULATE THE DIABETES MELLITUS RELATED VIOLATIONS OF PROLIFERATION, FUNCTIONS AND APOPTOSIS OF LEUKOCYTES

Authors:

Mariia Nagalievska (khmarija@gmail.com)
Mariya Sabadashka (m.sabadashka@meta.ua)
Halyna Hachkova (klevetag@gmail.com)
Nataliia Sybirna (sybirnanatalia@yahoo.com)

Version: 2 Date: 25 Apr 2017

Author’s response to reviews:

Dear editor,

The authors are sincerely grateful for the thorough work done in editing this article. Thank you for the analyzing the publication and offered the possibility to improve it.

In section “Preparation of Galega officinalis Extract and compounds identification” (Methods) was added the exact GC/MS method used for the identification of Galega officinalis extract compounds.

Results on the chemical composition of the extract are presented in the section “Results” and are discussed in the section “Discussion”.

The tables were removed from the main text of manuscript and at the end of the manuscript.

Yours faithfully, authors.